Author's response to reviews

Title: Expression of HIF-1alpha and VEGF in colorectal cancer: association with clinical outcomes and prognostic implications

Authors:

Dan Cao (caodan316@163.com)
Mei Hou (caodan316@163.com)
Song Yong Guan (guanyongsong@yahoo.com.cn)
Ming Jiang (mingminyu@sina.com)
Yu Yang (yangyuflying@hotmail.com)
Feng Hong Gou (joan.gou@gmail.com)

Version: 7 Date: 20 October 2009

Author's response to reviews: see over
Dear Editor,

Thank you very much for your letter and the comments from the referees to our paper submitted to BMC CANCER (Manuscript ID: 2007347659255943). We would like to express our appreciation for you and for the reviewers for the suggestions to improve the paper. The reviewers’ comments are very helpful for our work. We have discussed the comments carefully and have made revisions which we hope will meet your requirements. The written English is revised extensively with caution on correcting grammatical and typo errors with the help of a native English speaking scholar.

Early consideration of this revised manuscript at your earliest convenience is highly appreciated. Again, thank you for your patience and time.

Yours Sincerely,

M. D. Mei Hou,
Corresponding Author,

Department of Medical Oncology, Cancer Center of West China Hospital, Sichuan University, NO.37,Guoxue Road, Chengdu, Sichuan Province,610041,China
E-mail address:caodan316@163.com
Phone numbers: +86-028-85423609
Fax numbers: +86-028-85423609